Author's response to reviews

Title: The relation between smokeless tobacco and cancer in Northern Europe and North America. A commentary on differences between the conclusions reached by two recent reviews.

Authors:

Peter N Lee (PeterLee@pnlee.co.uk)
Jan Hamling (JanHamling@pnlee.co.uk)

Version: 2 Date: 18 February 2009

Author's response to reviews:

Following submission of the paper last week, we realized that the wording of the text could be improved in a number of places so have submitted a revision before the paper goes to the referees. There are no differences between the two versions in the basic details of the comparison we describe between our review of smokeless tobacco and cancer and that of Boffetta et al.